**Author's response to reviews**

**Title:** The anti-inflammatory and antioxidant activity of 25 plant species used traditionally to treat pain in southern African

**Authors:**

Salmon A Adebayo (adebayosa@tut.ac.za)
Jean P Dzoyem (jpdzoyem@yahoo.fr)
Leshweni J Shai (shailj@tut.ac.za)
Jacobus N Eloff (kobus.eloff@up.ac.za)

**Version:** 2  
**Date:** 9 September 2014

**Author's response to reviews:**

we have now added page numbers and line numbers as requested